Author’s response to reviews

Title: Mapping of health system functions to strengthen priority programs. The case of maternal health in Mexico.

Authors:

- Miguel Angel Gonzalez Block (miguel.gonzalezblock@insp.mx)
- Mariel V. Rouvier (mariel.rouvier@insp.mx)
- Víctor Becerril (victor.becerril@insp.mx)
- Paola M. Sesia (sesia@ciesas.edo.mx)

Version: 4 Date: 7 December 2010

Author’s response to reviews:

Dear BMC PH,

We are happy to submit our revised version according to Reviewer 2 comments, responding to all of them. I am attaching a point-by-point table as well as the revised draft. The draft indicates in yellow highlight all additions, while deletions are indicated in the accompanying table.